Author's response to reviews

Title: Monitoring the level of government trust, risk perception and intention of the general public to adopt protective measures during the influenza A (H1N1) pandemic in the Netherlands

Authors:

Willemien van der Weerd (willemien_van_der_weerd@hotmail.com)
Daniëlle RM Timmermans (drm.timmermans@vumc.nl)
Desirée JMA Beaujean (desiree.beaujean@rivm.nl)
Jurriaan Oudhoff (j.oudhoff@vumc.nl)
Jim E van Steenbergen (jim.van.steenbergen@rivm.nl)

Version: 5 Date: 13 July 2011

Author's response to reviews: see over
Editor BMC

Dear editor,

First we would like to thank you for accepting our manuscript, in principal, to BioMedCentral. Also, we would like to thank the editor team for the useful and helpful feedback we have received for formatting the manuscript to the journal style. This enabled us to further improve the manuscript.

Major revisions

We have written the list of authors in the manuscript exactly as they are in the submission system with respect to style and order.

The affiliations have been written in full, according to the recommended format: Department, Institute, (Street), City, (Postal code), Country.

Regarding the figure legends, we have removed all graphics from the legends (figure legend 1, 2, 4 and 5). Instead we mentioned them as ‘black line’ and ‘black bordered white circle’.

We have removed all vertical lines from the tables and inserted only horizontal lines.

Finally, the figures have been cropped as closely as possible.